Author's response to reviews

Title: Visualization of anomalous origin and course of coronary arteries in 748 consecutive symptomatic patients by 64-slice computed tomography angiography

Authors:

Franz von Ziegler (franz.ziegler@med.uni-muenchen.de)
Marco Pilla (marco.pilla@jax.ufl.edu)
Lori McMullan (lori.mcmullan@jax.ufl.edu)
Prasad Panse (prasad.panse@jax.ufl.edu)
Alexander W Leber (alexander.leber@med.uni-muenchen.de)
Norbert Wilke (norbert.wilke@yahoo.com)
Alexander Becker (alexander.becker@med.uni-muenchen.de)

Version: 3 Date: 2 December 2009

Author's response to reviews: see over
Dear Editor, dear BMC Editorial Production Team,

Thank you for accepting and editing the above cited manuscript. We appreciate your help to improve our manuscript and are delighted that you are willing to publish our work. We also hope that we now meet your formatting requirements.

Please find below the changes of the manuscript according to your suggestions.

1. Authors’ contributions: It is now indicated that all authors have read and approved the final manuscript.
2. Tables: We included them now in the manuscript in a tables section following the references. We also moved the table title to above the table and the legend to below the table.
3. Highlighting: All highlighting was removed.

Yours sincerely,

Franz von Ziegler, MD

---

Editor-in-Chief
BMC Cardiovascular Disorders

621 S New Ballas Rd
Ste 3016B
Saint Louis, MO, 63141

Melissa Leigh Norton, MD

12/02/2009

v. Ziegler
phone: +49 (89) 7095-2229; fax: +49 (89) 7095-2224